Author's response to reviews

Title: Subarachnoid hemorrhage secondary to a ruptured middle cerebral aneurysm in a patient with osteogenesis imperfecta: A case report

Authors:

Toshio Hirohata (thepolarbear523@gmail.com)
Satoru Miyawaki (miyawaki-tky@umin.ac.jp)
Akiko Mizutani (pochiamy@yahoo.co.jp)
Takayuki Iwakami (takayukiiwakami@yahoo.co.jp)
So Yamada (yamadaso0470@gmail.com)
Hajime Nishido (nishido-tky@umin.org)
Yasutaka Suzuki (yasuzuki-tky@umin.ac.jp)
Shinya Miyamoto (shinyamiyamoto@lily.ocn.ne.jp)
Katsumi Hoya (khoya@med.teikyo-u.ac.jp)
Mineko Murakami (muraminechan@yahoo.co.jp)
Akira Matsuno (akirakun@med.teikyo-u.ac.jp)

Version: 2
Date: 31 January 2014

Author's response to reviews:

Dr. Timothy Shipley
Executive editor
BMC Neurology

Dear Dr. Shipley:

I would like to thank you for your indication mail. I revised and resubmitted my manuscript titled “Subarachnoid hemorrhage secondary to a ruptured middle cerebral aneurysm in a patient with osteogenesis imperfecta: A case report” (MS: 5924958741199680). The revised points are following:

#1. Abstract: Please format your abstract according to the guidelines for authors: Background, Case presentation, Conclusion.

I corrected the paragraph construction of the abstract according to your instruction.

#2. Tables as additional files: If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references.

I arranged the table in the main manuscript following the references section.

Thank you very much again for considering our manuscript.
Best regards,

Toshio Hirohata, MD, Corresponding author
Department of Neurosurgery
Teikyo University Chiba Medical Center